Author's response to reviews

Title: Subregional 6-[18F]fluoro-L-m-tyrosine Uptake in the Striatum in Parkinson’s Disease

Authors:

Sayaka Asari (sasari@jichi.ac.jp)
Ken-ichi Fujimoto (kenchan@jichi.ac.jp)
Akihiro Miyauchi (miyauchi@webnettechnology.com)
Toshihiko Sato (toshihiko@doctor-net.co.jp)
Imaharu Nakano (inakano@jichi.ac.jp)
Shin-ichi Muramatsu (muramats@jichi.ac.jp)

Version: 3 Date: 17 March 2011

Author's response to reviews: see over
March 17, 2011

The BMC Series Editorial Production Team

BMC Neurology

Dear Sir:

Manuscript: 8229569746198738

Thank you for your suggestions. We have revised the manuscript.

Major revisions:
- We have provided figure titles.
- Table 1b is changed to Table 2.
  Table 2 and Table 3 are now changed to Table 3 and Table 4, respectively.
- Tables are included in the manuscript in a section after the references.

Minor revisions:
- Figures are cropped to minimize white space.
- We give color to Figure 3.
- Left square bracket is added for references [16-19] on page 9.
- A line feed is inserted after reference 28 on page 16.

We hope that the manuscript is now suitable for publication.

Thank you again for your help.

Yours sincerely,

Sayaka Asari, M.D.

Shin-ichi Muramatsu, M.D.

Division of Neurology, Department of Medicine, Jichi Medical University
3311-1 Yakushiji, Shimotsuke, Tochigi 3290498, Japan
Tel: +81-285-58-7352; Fax: +81-285-44-5118
E-mail: sasari@jichi.ac.jp or muramats@jichi.ac.jp